\begin{table}%t4
\caption{Photometric data for the nuclear region of NGC~1068 from high-resolution observations.}\label{tabobs1068}
\small%\centerline
{
\begin{tabular}{c c c c}
\hline\hline
Wavelength & Flux & Phot. aperture & Reference \\
$\rm\mu m$ & Jy & arcsec & \\ \hline
1.3 & $0.00048$ & 0.2 & R98 \\
1.6 & $0.07 \pm 0.02$ & 0.1 & W04 \\
1.8 & $0.0054$ & 0.2 & R98 \\
2.1 & $0.35 \pm 0.09$ & 0.1 & W04 \\
2.2 & $0.056$ & 0.27 & P04 \\
3.5 & $1.34$ & 0.4 & M00 \\
3.8 & $1.42$ & 0.4 & M03 \\
4.5 & $2.5$ & 0.27 & P04 \\
4.7 & $2.72 \pm 0.14$ & 0.4 & M03 \\
4.8 & $3.23 \pm 0.32$ & 0.4 & M00 \\
7.7 & $8.8 \pm 1.0$ & 0.4 & T01 \\
8.7 & $10.0 \pm 1.2$ & 0.4 & T01 \\
9.7 & $7.04 \pm 0.80$ & 0.4 & T01 \\
10.4 & $7.94 \pm 0.87$ & 0.4 & T01 \\
11.7 & $17.8 \pm 2.2$ & 0.4 & T01 \\
12.3 & $17.2 \pm 2.1$ & 0.4 & T01 \\
18.5 & $20.2 \pm 3.4$ & 0.4 & T01 \\ \hline
\end{tabular}}
\medskip
R98 \citep{Rou98}; W04 \citep{Wei04}; 
P04 \citep[Prieto, published in][]{Sch05}; 
M00 \citep{Mar00}; M03 \citep{Mar03}; 
T01 \citep{Tom01}.
\end{table}